Dear Editor,

Please find attached the revised manuscript ‘Measuring the health impact of human rights violations related to Australian asylum policies: a mixed methods study’ (Version 3).

The revisions made to this paper are as follows:

1. All footnotes have been deleted from the manuscript.
2. Vertical elements have been removed from all tables.
3. All references are now cited within the text in ascending numerical order.
4. All checked changes have been removed from the paper and there are no blank pages.

Yours sincerely,

Dr. Milica Markovic
Submitting author
Senior Research Fellow
School of Psychology, Psychiatry and Psychological Medicine
Monash University